Author's response to reviews

Title: Changes in the incidence of occupational disability as a result of back and neck pain in the Netherlands

Authors:
Ivan A Steenstra (i.a.steenstra@amc.uva.nl)
Jos H Verbeek (j.h.verbeek@amc.uva.nl)
Femmeke J Prinsze (.prinsze@erasmusmc.nl)
Dirk L Knol (d.knol@vumc.nl)

Version: Date: 5 July 2006

Author's response to reviews:

Dear Iraxte,

You have send me an e-mail with regards of the medical ethics comitee approval.

Because of the hollidays and me moving to Canada it has taken me som etim eto make the necessary alterations.

I have added the following text on page 6:

Since people were not subjected to procedures or were required to follow rules of behavior, an approval of the Medical Ethics Committee of the Academic Medical Centre was not required according to the Medical Research Involving Human Subjects Act (WMO).

I hope this will suffice. We consulted the MEC of AMC and they told us that we didn't have to follow the normal procedures bas on WMO.

With kind regards,

Ivan
Ivan Steenstra
Institute for Work & Health
481 University Ave,
Ste 800
Toronto, Ontario
M5G 2E9, Canada
Tel: 416-927-2027 ext. 2155
Fax: 416-927-4167
E-mail: isteenstra@iwh.on.ca
Website: www.iwh.on.ca